



# **Geophysical Research Letters**

## **RESEARCH LETTER**

10.1002/2018GL077353

#### **Key Points:**

- Source distribution and seasonal variation of size-aggregated sulfate aerosols collected at Baring Head, New Zealand were investigated
- Coarse aerosol sulfate was mainly from sea salt; sulfate in fine aerosols was mostly secondary sulfate
- Most secondary sulfate was originated from marine DMS emission and should explain the seasonal variations of Southern Ocean CCN

**Supporting Information:** 

• Supporting Information S1

#### Correspondence to: J. Li,

li2502@purdue.edu

#### Citation:

Li, J., Michalski, G., Davy, P., Harvey, M., Katzman, T. & Wilkins, B. (2018). Investigating source contributions of size-aggregated aerosols collected in Southern Ocean and Baring Head, New Zealand using sulfur isotopes. *Geophysical Research Letters*, *45*, 3717–3727. https://doi.org/10.1002/ 2018GL077353

Received 29 JAN 2018 Accepted 24 MAR 2018 Accepted article online 30 MAR 2018 Published online 16 APR 2018

#### ©2018. American Geophysical Union. All Rights Reserved.

## Investigating Source Contributions of Size-Aggregated Aerosols Collected in Southern Ocean and Baring Head, New Zealand Using Sulfur Isotopes

Jianghanyang Li<sup>1</sup> , Greg Michalski<sup>1,2</sup>, Perry Davy<sup>3</sup>, Mike Harvey<sup>4</sup>, Tanya Katzman<sup>2</sup>, and Benjamin Wilkins<sup>2</sup>

<sup>1</sup>Department of Earth, Atmospheric, and Planetary Sciences, Purdue University, West Lafayette, IN, USA, <sup>2</sup>Department of Chemistry, Purdue University, West Lafayette, IN, USA, <sup>3</sup>National Isotope Centre, Institute of Geological and Nuclear Sciences, Lower Hutt, New Zealand, <sup>4</sup>National Institute of Water and Atmospheric Research, Wellington, New Zealand

**Abstract** Marine sulfate aerosols in the Southern Ocean are critical to the global radiation balance, yet the sources of sulfate and their seasonal variations are unclear. We separately sampled marine and ambient aerosols at Baring Head, New Zealand for 1 year using two collectors and evaluated the sources of sulfate in coarse (1–10 µm) and fine (0.05–1 µm) aerosols using sulfur isotopes ( $\delta^{34}$ S). In both collectors, sea-salt sulfate (SO<sub>4</sub><sup>2–</sup><sub>SS</sub>) mainly existed in coarse aerosols and nonsea-salt sulfate (SO<sub>4</sub><sup>2–</sup><sub>NSS</sub>) dominated the sulfate in fine aerosols, although some summer SO<sub>4</sub><sup>2–</sup><sub>NSS</sub> appeared in coarse particles due to aerosol coagulation. SO<sub>4</sub><sup>2–</sup><sub>NSS</sub> in the marine aerosols was mainly (88–100%) from marine biogenic dimethylsulfide (DMS) emission, while the SO<sub>4</sub><sup>2–</sup><sub>NSS</sub> in the ambient aerosols was a combination of DMS (73–79%) and SO<sub>2</sub> emissions from shipping activities (~21–27%). The seasonal variations of SO<sub>4</sub><sup>2–</sup><sub>NSS</sub> concentrations inferred from the  $\delta^{34}$ S values in both collectors were mainly controlled by the DMS flux.

**Plain Language Summary** Marine sulfate aerosols are critical to the global radiation balance through directly scattering sunlight or forming clouds; however, their feedback effects are poorly quantified because their sources and size distributions are unclear. We investigated the origins and size distributions of sulfate aerosols from the Southern Ocean as well as the ambient environment at Baring Head, New Zealand. We found that the sulfate in coarse (>1 um) aerosols was dominated by sea-salt sulfate; while the fine aerosol (<1 um) sulfate, which could act as cloud condensation nuclei, was mostly formed via atmospheric oxidation of sulfur-bearing gases. The origin of the secondary sulfate was then identified using sulfur isotopic analysis: Dimethylsulfide emitted by phytoplankton contributed over 90% of the secondary sulfate on the Southern Ocean; it also contributed ~73–79% of secondary sulfate aerosols in the ambient air at Baring Head, while the remainder was from anthropogenic sulfur emissions. Our work suggest that marine biological activity is an important factor that controls the amount of sulfate aerosols in remote marine atmosphere, which is of great importance to global climate models.

## **1. Introduction**

Sulfate is one of the major inorganic components in aerosols and is essential to aerosol nucleation and accumulation processes (Andronache et al., 1997; Kulmala et al., 2000), which greatly affect Earth's radiation budget. However, the impact of sulfate aerosols to the radiation budget is still poorly quantified (IPCC, 2007), partly because the sources and fluxes of sulfate show strong spatial and seasonal variations. In the marine boundary layer, sulfate aerosols consist of primary and secondary sulfate. Primary sulfate, that is, sea-salt sulfate (SO<sub>4</sub><sup>2–</sup><sub>SS</sub>) is formed via sea spray and air bubble bursting at the ocean surface (Lewis & Schwartz, 2004; O'Dowd et al., 2007); its size distribution and flux are controlled by wind speed, wave height, and other meteorological conditions (Gong, 2003; Lewis & Schwartz, 2004; Van Eijk et al., 2011). Nonsea-salt sulfate (SO<sub>4</sub><sup>2–</sup><sub>NSS</sub>), also called secondary sulfate, is formed via atmospheric oxidation of S, the sources of which are (1) dimethylsulfide (SO<sub>4</sub><sup>2–</sup><sub>DMS</sub>) emitted by marine phytoplankton (Barnes et al., 2006); (2) natural terrestrial S emission (SO<sub>4</sub><sup>2–</sup><sub>nat</sub>), including sulfate from SO<sub>2</sub> emitted by volcanic and terrestrial biological activities; or (3) anthropogenic S emissions (SO<sub>4</sub><sup>2–</sup><sub>anth</sub>). In the midlatitudes of the Southern Ocean (~35–45°S), SO<sub>4</sub><sup>2–</sup><sub>DMS</sub> is thought to play a critical role in the cloud physics and climate by controlling the number of cloud condensation nuclei (CCN), resulting in strong reflection of solar radiation in the summer when the DMS flux is high (Ayers & Gras, 1991; Boers et al., 1994; Korhonen et al., 2008; McCoy et al., 2015). However, the sources, size

distributions, and seasonal variations of the Southern Ocean sulfate are not well understood, which limits our ability to predict its response to the changing climate.

Previous studies have investigated the sources and size distributions of marine sulfate aerosols in several locations (Calhoun & Bates, 1989; Calhoun et al., 1991; Faloona, 2009; Ghahremaninezhad et al., 2016; Norman et al., 1999; Novák et al., 2001; Rempillo et al., 2011; Seguin et al., 2011); however, several questions remain unanswered. First, substantial amounts of  $SO_4^{2-}_{NSS}$  had been observed in the coarse aerosols (0.9–16 µm) sampled at Baring Head originating from the biologically productive subtropical frontal region of the Chatham Rise (Sievering et al., 2004), suggesting  $SO_4^{2-}_{NSS}$  was mainly formed on coarse sea-salt particles. This is in contrast with observations in the Northern Hemisphere (Ghahremaninezhad et al., 2016; Norman et al., 1999; Rempillo et al., 2011; Seguin et al., 2011) and modeling results (Alexander et al., 2005), all of which suggested  $SO_4^{2-}_{NSS}$  should mainly distributed in the fine particles (<0.95 µm). Second, the contribution of anthropogenic emission to the Southern Ocean sulfate is uncertain. Capaldo et al. (1999) suggested that international shipping emission, which is the main anthropogenic sulfur source in midlatitude Southern Ocean (~40°S), contributes to ~5% to 20% of total sulfate, while McCoy et al. (2015) showed a higher anthropogenic contribution of 20%–35% at 40°S. Third, the observed seasonal variations of Southern Ocean sulfate (e.g., McCoy et al., 2015; Udisti et al., 2012) were unexplained and difficult to predict because the variation of each sulfate source was unknown.

Geochemical and sulfur isotopic analyses are useful tools in determining the sources of sulfate. Cation and anion analyses are useful in partitioning between  $SO_4^{2-}_{SS}$  and  $SO_4^{2-}_{NSS}$ , and changes in the sulfur isotopic composition ( $\delta^{34}S$ ) can help in differentiating  $SO_4^{2-}$  sources:  $\delta^{34}S$  of  $SO_4^{2-}_{SS}$  ( $\delta^{34}S_{SS}$ ) is a constant +21‰ (Rees et al., 1978),  $\delta^{34}S$  values of DMS range between +15‰ and +19‰ (Amrani et al., 2013; Krouse & Grinenko, 1991; Oduro et al., 2012), while most anthropogenic and terrestrial sulfates display much lower  $\delta^{34}S$  values between -5% and +10% (Calhoun et al., 1991; Krouse & Grinenko, 1991; McArdle et al., 1998; Sakai et al., 1982; Zhu et al., 2016). In this work, a full-year sampling campaign was conducted to collect size-segregated aerosols derived from the marine sector (subtropical frontal region and the Southern Ocean) and all sectors (ambient environment) at Baring Head, New Zealand (41.4°S, 174.9°E, Figure 1). Geochemical and sulfur isotopic analyses were used to investigate S sources and their seasonal variations.

## 2. Method

The aerosol sampling campaign spanned from 30 June 2015 to 11 August 2016 and sampled both marine and ambient air. Two high-volume samplers equipped with Cascade Impactors were set up on a 15 m tower at Baring Head, New Zealand (Figure 1), where the metrological conditions were measured, to collect coarse  $(1-10 \,\mu\text{m})$  and fine aerosols  $(0.05-1.0 \,\mu\text{m})$  at a flow rate of  $\sim 1 \,\text{m}^3/\text{min}$ . Cellulose filters were used to minimize ion contamination (especially Na<sup>+</sup> and SO<sub>4</sub><sup>2-</sup>), which were tested using eight blanks and resulted in  $[Na^+] < 0.6\%$  of average  $[Na^+]$  in the field samples (e.g., < 0.05 nmol/m<sup>3</sup>). One of the collectors (marine collector, set on top of the tower) was configured to sample oceanic air masses by only collecting when the average southerly (onshore) wind speed was above 5 m/s for at least 30 min. Filters on this collector were replaced after 100 to 150 hr of collection to acquire enough sample (every 2 to 6 weeks). The ambient collector sampled air mass regardless of the wind direction, and filters were replaced every 7 days (154 hr). Previous studies (Dorling et al., 1992; Steinkamp et al., 2017) confirmed the origins of the collected aerosols using cluster analysis (Figure 1) showing aerosols in the marine collector originated from the oceanic sector of the subtropical front/Chatham Rise (red shaded area), while the aerosols in the ambient collector were from both the ocean and the New Zealand mainland (blue lines). Meantime, a wave buoy off the coast was used to measure the wave peak height. The filters were carefully sealed and shipped to Purdue Stable Isotope Laboratory, where standard methods were used to analyze cations (inductively coupled plasma optical emission spectrometry, ICP-OES), anions (Ion Chromatography), and sulfur isotopic compositions of sulfate (isotope-ratio mass spectrometry, see supporting information for detailed description of analytical methods and data). The concentrations of  $SO_4^{2-}S_5$  were calculated using the seawater  $[SO_4^{2-}Na^+]$  molar ratio (0.058) and the measured Na concentrations:  $[SO_4^{2-}S_5] = 0.058*[Na^+]$  (Keene et al., 2007). The residual sulfate was considered  $SO_4^{2-}NS_5$ . This assumes sodium is a conservative element, seawater is its sole source, and it is insensitive to chemical or biological loss (Keene et al., 1986).







## 3. Results and Discussions

#### 3.1. Distribution of Sea-Salt and Nonsea-Salt Sulfate in Coarse and Fine Aerosols

Sulfate in coarse aerosols was mainly  $SO_4^{2-}_{SS}$ , and  $SO_4^{2-}_{NSS}$  was found primarily in the fine aerosol fraction, which was in contrast with a previous study at Baring Head (Sievering et al., 2004). Coarse aerosol sulfate concentrations in the two collectors were similar ( $8.8 \pm 1.9 \text{ nmol/m}^3$  in marine collector and  $8.1 \pm 3.8 \text{ nmol/m}^3$  in ambient collector), which is remarkably close to a previous study at Baring Head that averaged 8.3 nmol/m<sup>3</sup> (Sievering et al., 2004). SO<sub>4</sub><sup>2-</sup><sub>SS</sub> accounted for  $88 \pm 5\%$  of the coarse aerosol sulfate (Figures 2a and 2b), but the contribution of  $SO_4^{2-}_{NSS}$  in the course mode slightly increased during the summer (discussed below). In contrast, fine particle sulfate concentrations were lower, (averaging 1.7 nmol/m<sup>3</sup> in marine collector and



**Figure 2.** (a–d)  $SO_4^{2-}_{SS}$  (black) and  $SO_4^{2-}_{NSS}$  (red) concentrations in coarse and fine aerosols from ambient and marine collectors; blue lines in Figures 2a and 2c are averaged (to fit the marine collector sampling window) total sulfate concentrations. (e) Wind speed, wave height, and coarse  $SO_4^{2-}_{SS}$  concentrations in ambient collector.

2.7 nmol/m<sup>3</sup> in ambient collector), and SO<sub>4</sub><sup>2-</sup><sub>SS</sub> accounted for only ~16 ± 1% of total sulfate, the remaining 84 ± 1% was SO<sub>4</sub><sup>2-</sup><sub>NSS</sub>. This general SO<sub>4</sub><sup>2-</sup><sub>SS</sub> distribution pattern was similar to the distribution previously observed at Baring Head by Sievering et al. (2004), but in that study only 53% of coarse aerosol sulfate was attributed to SO<sub>4</sub><sup>2-</sup><sub>SS</sub> compared to this study's 88%. They also suggested coarse aerosol sulfate accounted for 81% of total SO<sub>4</sub><sup>2-</sup><sub>NSS</sub>, while our results showed a much lower contribution of ~24–36%.

The discrepancy in  $SO_4^{2-}_{NSS}$  distribution could be attributed to either (1) differences in sampling conditions or (2) differences in  $SO_4^{2-}_{SS}$  estimation. Sievering et al. (2004) suggested that under clear-sky conditions, pH buffering by excess calcium provided from coccolithophores fragments could enhance the effectiveness of ozone-mediated oxidation in sea-salt aerosol droplets, leading to greater coarse mode  $SO_4^{2-}_{NSS}$ . In this work, aerosols from both clear and cloudy conditions were sampled; therefore, this mechanism was likely less important and results in lower coarse mode  $SO_4^{2-}_{NSS}$ . Alternatively, the discrepancy may be due to our use of aerosol [Na<sup>+</sup>] to estimate  $SO_4^{2-}_{SS}$  rather than [Mg<sup>2+</sup>] used by Sievering et al. (2004). Sodium is the most widely used conservative tracer of sea salt (Lin et al., 2017; Norman et al., 1999; Udisti et al., 2016) because (1) it has a high concentration in the seawater, (2) nonmarine sodium inputs are trivial in most places (McInnes et al., 1994), and (3) it is insensitive to secondary alterations. Sievering et al. (2004) used [Mg<sup>2+</sup>] to calculate  $SO_4^{2-}_{SS}$  fraction because of abnormally high  $[Na^+Cl^-]$  and  $[Na^+Mg^{2+}]$  ratios in their aerosols that they attributed to either contamination by filter blanks or possible contributions of sodium in soil. Likewise,  $[Mg^{2+}]$  has been used to estimate  $SO_4^{2-}_{SS}$  in aerosols collected at Cape Grim, Tasmania, and La Jolla, California because sodium from either filter blanks or soil was believed to be significant (Cainey et al., 1999; Hill-Falkenthal et al., 2012; Priyadarshi et al., 2012). The cellulose filters used in this study had a very low Na<sup>+</sup> blank relative to typical glass fiber filters (Dams et al., 1972), and Southern Ocean derived air masses should have minimal terrestrial influence at Baring Head. Further, the coarse aerosol  $[Mg^{2+}Na^+]$  in both of our collectors averaged at 0.111 ± 0.005, much higher than the 0.05 ratio determined by Sievering et al. (2004) and very close to the seawater ratio of 0.11 (Keene et al., 2007), indicating minimal contribution of sodium from either soil or filter blank. Furthermore, the low  $[Mg^{2+}Na^+]$  ratio observed by Sievering et al. (2004) may be a consequence of biased  $[Mg^{2+}]$  measurements. A recent study has observed that sea-spray  $Mg^{2+}$  and  $Ca^{2+}$  form complexes with lipids, fatty acids, and saccharides (Jayarathne et al., 2016) during sea-salt aerosol formation, potentially biasing  $[Mg^{2+}]$  analyzed by ion chromatography (Sievering et al., 2004). Using ICP-OES to analyze  $[Mg^{2+}]$  does not require it to be in ionic form, so could be a more accurate measure of  $[Mg^{2+}Na^+]$  in the sea-salt aerosols given the possibility of complexation and provides confidence in the accuracy of the  $SO_4^{2-}_{SS}$  presented here.

The size distribution patterns of  $SO_4^{2-}_{SS}$  and  $SO_4^{2-}_{NSS}$  agrees with other field observations and atmospheric modeling results, indicating  $SO_4^{2-}_{NSS}$  is the main sulfur source in CCN. Observations in the North Atlantic, Arctic, and Pacific Ocean have suggested similar distribution patterns, of which >90% of  $SO_4^{2-}_{SS}$  was distributed in coarse aerosols and > 60% of  $SO_4^{2-}_{NSS}$  was in fine aerosols (Ghahremaninezhad et al., 2016; Murphy et al., 1998; Norman et al., 1999; Rempillo et al., 2011; Seguin et al., 2011). Therefore, we suggest in the Southern Ocean  $SO_4^{2-}_{NSS}$  was also the main sulfate source of CCN, since CCN are usually <0.2  $\mu$ m (Hudson & Noble, 2006). Additionally, we noticed coarse aerosol  $SO_4^{2-}_{NSS}$  was observed when the total  $SO_4^{2-}_{NSS}$  concentrations ( $SO_4^{2-}_{NSS-coarse} + SO_4^{2-}_{NSS-fine}$ ) exceeded 2.0 nmol/m<sup>3</sup>. This indicates higher  $SO_4^{2-}_{NSS}$  concentrations helped coagulation forms larger  $SO_4^{2-}_{NSS}$  particles. Further experiments and field sample analysis should take place to quantify the relationship between  $SO_4^{2-}_{NSS}$  concentration and their size distribution.

The variations in  $SO_4^{2-}_{SS}$  concentrations were attributed to variations in wind speed and wave heights at Baring Head. Early studies have suggested that the mass concentration of sea-salt aerosols is positively correlated to wind speed (O'Dowd et al., 1997; O'Dowd & Smith, 1993); however, recent studies had suggested other physical conditions, such as whitecap coverage and wave peak height on the coast are also important (Clarke et al., 2006; Lewis & Schwartz, 2004; Mårtensson et al., 2003; O'Dowd et al., 2007). Our observed  $SO_4^{2-}_{SS}$ concentrations only showed a weak correlation with the average wind speed (P > 0.05,  $R^2 = 0.03$ , Figure 2e) similar to other field studies (Ghahremaninezhad et al., 2016; Jaeglé et al., 2011; Lewis & Schwartz, 2004; Rempillo et al., 2011; Seguin et al., 2011). In contrast, maximums in wave peak heights usually corresponded to high  $SO_4^{2-}_{SS}$  concentrations at Baring Head (Figure 2e), supporting the hypothesis that breaking waves increase sea-salt aerosol formation near the coast (Jensen et al., 1997; Monahan et al., 1986; Van Eijk et al., 2011). Therefore, we suggest that wave height was more important than wind speed, under low median wind speeds (9.9 ± 3.9 m/s), in generating sea-salt aerosols at Baring Head during the study period.

## 3.2. Sources and Seasonal Variations of SO<sub>4</sub><sup>2-</sup><sub>NSS</sub> in the Marine Collector

The sulfur isotopic compositions of sulfate also indicated a mixture of  $SO_4^{2-}_{SS}$  and  $SO_4^{2-}_{NSS}$ . The  $\delta^{34}S$  values of total sulfate ( $SO_4^{2-}_{SS} + SO_4^{2-}_{NSS}$ ) were interpreted using a two-end-member isotope mixing model (Figure 3a):  $SO_4^{2-}_{SS}$  with  $\delta^{34}S$  value that is the same (+21‰) as seawater (Rees et al., 1978) and  $SO_4^{2-}_{NSS}$  with lower  $\delta^{34}S$  value(s). Most coarse aerosol sulfate had  $\delta^{34}S$  values near the seawater end-member, indicating the coarse aerosol sulfate was predominately composed of  $SO_4^{2-}_{SS}$ , which supports our  $SO_4^{2-}_{SS}$  estimates using [Na<sup>+</sup>]. In contrast, the fine aerosol had lower  $\delta^{34}S$  values (+11‰ to +21‰), suggesting a higher proportion of  $SO_4^{2-}_{NSS}$  relative to the coarse aerosols.

The  $\delta^{34}$ S values of SO<sub>4</sub><sup>2-</sup><sub>NSS</sub> ( $\delta^{34}$ S<sub>NSS</sub>) can give insight into the origin of SO<sub>4</sub><sup>2-</sup><sub>NSS</sub> sources. The  $\delta^{34}$ S<sub>NSS</sub> values were calculated using  $\delta^{34}$ S<sub>NSS</sub> = ( $\delta^{34}$ S<sub>bulk</sub> - (1 - SO<sub>4</sub><sup>2-</sup><sub>NSS</sub>%)\* + 21%)/SO<sub>4</sub><sup>2-</sup><sub>NSS</sub>%, where SO<sub>4</sub><sup>2-</sup><sub>NSS</sub>% is the fraction of SO<sub>4</sub><sup>2-</sup><sub>NSS</sub> (using [Na<sup>+</sup>]), the +21% is the  $\delta^{34}$ S value of seawater sulfate, and  $\delta^{34}$ S<sub>bulk</sub> is the  $\delta^{34}$ S value of the total aerosol sulfate. The calculated  $\delta^{34}$ S<sub>NSS</sub> values ranged from +6% to +19% (Figure 3).  $\delta^{34}$ S<sub>NSS</sub> values in the marine collector range from +15% to +19% (Figure 3b), which is the same range previously observed



**Figure 3.** (a)  $\delta^{34}$ S distribution of all samples, showing a mixing of SO<sub>4</sub><sup>2-</sup><sub>SS</sub> and SO<sub>4</sub><sup>2-</sup><sub>NSS</sub>; (b) the  $\delta^{34}$ S<sub>NSS</sub> values of ambient and marine fine sulfate; horizontal bars indicate the  $\delta^{34}$ S values of each end-member; and (c) estimated anthropogenic and DMS contribution to SO<sub>4</sub><sup>2-</sup><sub>NSS</sub> in ambient fine sulfate.

in  $\delta^{34}S_{DMS}$  (Krouse & Grinenko, 1991; Patris et al., 2002; Sanusi et al., 2006; Wadleigh, 2004). This indicates that SO<sub>4</sub><sup>2-</sup><sub>DMS</sub> was the dominant source of  $SO_4^{2-}_{NSS}$  in the marine collector. The ambient aerosol  $\delta^{34}S_{NSS}$ values were as low as +6‰, which cannot be explained by oxidation of DMS, and points to nonmarine sulfur sources. The main anthropogenic S sources in the region are international and local shipping activity (Capaldo et al., 1999; McCoy et al., 2015) and the  $\delta^{34}$ S value of sulfur derived from ships has been estimated to be +3‰ (Patris et al., 2000; Rempillo et al., 2011). Likewise, the  $\delta^{34}$ S values of sulfate derived from terrestrial sulfur sources near Baring Head also have been estimated to be +3‰ (discussed below). Thus, the range of Baring Head  $SO_4^{2-}NSS$  values is interpreted as a second mixture of  $SO_4^{2-}DMS$ (Calhoun et al., 1991; Krouse & Grinenko, 1991; Patris et al., 2002; Sanusi et al., 2006) and  $SO_4{}^{2-}_{anth}$  whose fractions ( $f_{DMS}$ ,  $f_{anth}$ ) can be quantified using a simple two end-member isotope mixing model:  $\delta^{34}S_{NSS} = f_{DMS}*\delta^{34}S_{DMS} + f_{anth}*\delta^{34}S_{anth}$ . Using this mixing model, we can first estimate the  $f_{\text{DMS}}$  and  $f_{\text{anth}}$  (given in %) in the marine collector. The high  $\delta^{34}S_{NSS}$  values (Figure 2b) of marine  $SO_4^{2-}_{NSS}$  indicate that it was primarily (88–100%)  $SO_4^{2-}DMS$  and only 0–12% was  $SO_4^{2-}anth$ . To estimate the upper limit of  $f_{anth}$ , we assume a constant  $\delta^{34}S_{DMS}$  endmember of +19‰ (Wadleigh, 2004), and the  $\delta^{34}S_{anth}$  end-member was from ship emissions with  $\delta^{34}S = +3\%$ . The variation of  $\delta^{34}S_{NSS}$ can be solely attributed to changing in  $f_{anth}$ , and the calculated upper limit of  $f_{anth}$  is ~12%. This SO<sub>4</sub><sup>2-</sup><sub>anth</sub> contribution to SO<sub>4</sub><sup>2-</sup><sub>NSS</sub> is much lower than the 20-35% estimated by McCoy et al. (2015) but in agreement with the 5-20% estimated by Capaldo et al. (1999). However, this calculation assumed the  $\delta^{34}S_{\text{DMS}}$  was a constant +19‰ and ignored the observed +15‰ to +19‰ range in other studies (Krouse & Grinenko, 1991; Patris et al., 2002; Sanusi et al., 2006; Wadleigh, 2004). This 4‰ variation in  $\delta^{34}S_{DMS}$  values is thought to be caused by a combination of (1) variation of isotopic fractionation during SO<sub>2</sub> oxidized to sulfate ( $\delta^{34}S_{sulfate}-\delta^{34}S_{SO2}$ , Harris et al., 2012, 2013) and (2) the  $\delta^{34}S$  of DMS gas could slightly vary (Amrani et al., 2013). If we consider the  $\delta^{34}S_{DMS}$  variation, then the contribution of anthropogenic S in marine aerosols could be as low as zero at Baring Head. This 0-12% anthropogenic S contribution to the SO<sub>4</sub><sup>2-</sup><sub>NSS</sub> budget is significantly lower than the estimates of 30-70% in the Northern Atlantic Ocean and other Pacific Ocean sites (Capaldo et al., 1999; Patris et al., 2000; Yang et al.,

2017). We suggest that this lower anthropogenic sulfate fraction in  $SO_4^{2-}_{NSS}$  at Baring Head is due to significantly lower sulfur emission in the Southern Hemisphere (IEA, 2014), fewer shipping routes in the Southern Hemisphere, thus minimal influence of ship sulfur emissions

(Paxian et al., 2010), and poor interhemispheric mixing of sulfate that minimizes the influence of the Northern Hemisphere sulfur emissions at Baring Head (Capaldo et al., 1999).

The observed seasonal variation of  $SO_4^{2-}_{NSS}$  in the marine collector (Figure 2d), which was primarily from DMS emissions, must be controlled by seasonal changes in the DMS flux. Baring Head's average summer  $SO_4^{2-}_{DMS}$  concentration was ~2.7 nmol/m<sup>3</sup>, while winter concentrations averaged at 0.6 nmol/m<sup>3</sup> (Figure 2d), similar to the  $SO_4^{2-}_{DMS}$  concentrations observed in Arctic, Southern Pacific, and Northern Atlantic Ocean (Allen et al., 1997; Ghahremaninezhad et al., 2016; Huang et al., 2017; Quinn et al., 2009; Rempillo et al., 2011; Seguin et al., 2011). The observed average summer  $SO_4^{2-}_{DMS}$  concentration was ~4.5 times higher than the winter  $SO_4^{2-}_{DMS}$ , which agrees with the recent observation of dissolved sea-water DMS concentration by Lana et al. (2011) who showed at least a fourfold higher concentration of dissolved

DMS in the austral summer compared to winter. In addition, Law et al. (2017) discussed evidence suggesting this seasonal variation presented by Lana et al. (2011) also occurred in the waters offshore of New Zealand. Therefore, since  $SO_4^{2-}_{DMS}$  was the main source of fine sulfate particles, the seasonal variation of  $SO_4^{2-}_{DMS}$  could explain the 300% increase of CCN in the summer observed by Ayers and Gras (1991).

#### 3.3. Sources and Seasonal Variations of SO<sub>4</sub><sup>2-</sup><sub>NSS</sub> in the Ambient Collector

The ambient aerosol  $\delta^{34}S_{NSS}$  values were lower than the marine aerosol  $\delta^{34}S_{NSS}$  and displayed a distinct seasonal trend, indicating changing contribution of anthropogenic sulfur over the year (Figure 3b). The ambient aerosol  $\delta^{34}S_{NSS}$  values (+6% to +18%) were close to those of marine aerosols during the summer, suggesting the  $SO_4{}^{2-}_{NSS}$  was also dominated by DMS during the summer months. In contrast, the  $\delta^{34}S_{NSS}$  values in the winter were significantly lower (Figure 3b) than in the marine aerosols, suggesting  $SO_4^{2-}_{nat}$  and/or SO<sub>4</sub><sup>2-</sup><sub>anth</sub> input (e.g., Calhoun et al., 1991; Ghahremaninezhad et al., 2016; Patris et al., 2000; Rempillo et al., 2011). Potential terrestrial sulfur sources include terrestrial biogenic emissions, volcanic emissions, and anthropogenic emissions. Terrestrial biogenic sulfur should be a minor contributor because the estimated biogenic sulfur flux ( $<1 \times 10^5$  mol/day) in New Zealand (Bates et al., 1992) is small compared to  $1 \times 10^6$  mol/day anthropogenic emission from Wellington, which is adjacent to the Baring Head site (Ministry of the Environment, 2004). The volcanic activity should also be minor at Baring Head since the nearest volcano and geothermally active regions with  $H_2S$  emission are ~300 km away, and there was no significant volcanic activity during our sampling period (GeoNet volcanic emission database). Therefore, SO<sub>4</sub><sup>2-</sup><sub>nat</sub> is likely small, and  $SO_4^{2-}_{anth}$  was likely the main nonmarine sulfur sources of  $SO_4^{2-}_{NSS}$  in the ambient collector. Eighty-two percent of anthropogenic sulfur emissions in the Wellington region were from commercial shipping (Ministry of the Environment, 2004) that uses low-grade fuel oil. Early study showed that the  $\delta^{34}$ S values of fuel oil center around 5‰ (Nielsen, 1974); recent measurements of ship emissions also have  $\delta^{34}$ S value of +3‰ ± 3‰ (Patris et al., 2000; Rempillo et al., 2011). Other anthropogenic sources, such as coal burning and industrial fossil fuel combustions, also showed  $\delta^{34}$ S values of +3‰ ± 3‰ (e.g., Górka et al., 2017; Proemse & Mayer, 2012). Rainwater sulfate collected at Gracefield, a semi-industrial district ~25 km inland from Baring Head and Wellington, had  $\delta^{34}$ S values of ~ +3 ± 5‰ (Mizutani & Rafter, 1969), which was consistent with the end-members. Thus, the ambient aerosol  $\delta^{34}S_{anth}$  end-member was estimated to be ~ +3‰ and the  $\delta^{34}S_{NSS}$  in the ambient collector can be again interpreted as a mixing between  $SO_4^{2-}_{anth}$  and  $SO_4^{2-}_{DMS}$  using same isotope mixing model (above).

The  $f_{anth}$  values varied within and by season: ~40–60% in the winter, ~20–40% in the spring and fall, and 0–20% in the summer. The SO<sub>4</sub><sup>2-</sup><sub>anth</sub> concentrations ( $f_{anth}*[SO_4^{2-}_{NSS}]$ ) range from 0 to 2.5 nmol/m<sup>3</sup>, with an annual average of 0.6 nmol/m<sup>3</sup>, accounted for 27% of total SO<sub>4</sub><sup>2-</sup><sub>NSS</sub>. Then, assuming the SO<sub>4</sub><sup>2-</sup><sub>NSS</sub> in the marine collector was purely derived from DMS, the  $\delta^{34}S_{NSS}$  in the marine collector can be used as the  $\delta^{34}S_{DMS}$  in the equation. Under this situation, the  $f_{anth}$  values were ~40–50% in the winter, ~5–20% in the spring and fall, and 0–15% in the summer (Figure 3c). The annual average  $f_{anth}$  was 21%. Therefore, we suggest the SO<sub>4</sub><sup>2-</sup><sub>anth</sub> accounted for 21–27% of total SO<sub>4</sub><sup>2-</sup><sub>NSS</sub> at Baring Head. Davy (2007) and Davy et al. (2008, 2012) had also observed similar SO<sub>4</sub><sup>2-</sup><sub>NSS</sub> concentrations and seasonal variations at Lower Hutt, a coastal town ~20 km from Baring Head, but the contribution of SO<sub>4</sub><sup>2-</sup><sub>DMS</sub> was unexamined. We suggest their SO<sub>4</sub><sup>2-</sup><sub>NSS</sub> sources were similar to our results: SO<sub>4</sub><sup>2-</sup><sub>DMS</sub> contributed >70% of total SO<sub>4</sub><sup>2-</sup><sub>NSS</sub>, and SO<sub>4</sub><sup>2-</sup><sub>anth</sub> accounted for the rest.

The anthropogenic sulfate contributions at Baring Head were significantly lower than the Northern Hemisphere marine boundary layer. The 21–27% anthropogenic contribution was much lower than most studies in the Northern Hemisphere coastal areas where 30% to >90%  $SO_4^{2-}_{NSS}$  was anthropogenic (Ghahremaninezhad et al., 2016; Seguin et al., 2011; Xu & Gao, 2015) but only slightly higher than remote Midway Island (20%) in the northern Pacific Ocean (Savoie & Prospero, 1989). Local sulfur emission at Midway Island is minor (IEA, 2014), thus  $SO_4^{2-}_{anth}$  should represent the sulfate background in the Northern Hemisphere. In contrast, Baring Head is ~20 km from Wellington that has significant SO<sub>2</sub> emission, and our data suggest this is the major, if not sole,  $SO_4^{2-}_{anth}$  source at Baring Head. This means the background  $SO_4^{2-}_{anth}$  level near the Southern Ocean must be very low, in agreement with our observation from the marine collector and the modeled results in Capaldo et al. (1999). Future atmospheric modeling work incorporated with sulfur isotopic module could potentially distinguish the Southern Hemisphere background  $SO_4^{2-}_{anth}$  from  $SO_4^{2-}_{anth}$  emissions from Wellington.

 $SO_4^{2-}_{DMS}$  collected in the ambient aerosols showed similar seasonal trend as the marine collector, while the anthropogenic sulfate displayed little seasonal variation (Figure 3c). Sulfate source apportionment determined using sulfur isotopes allows us to examine the seasonal variations of both DMS and anthropogenic derived sulfate. Similar to the  $SO_4^{2-}$  from the marine collector, the ambient  $SO_4^{2-}_{DMS}$  shows lower flux in the winter and higher flux in the summer but with a more significant variation (0.5 to 4 nmol/m<sup>3</sup>) throughout the year. This is because the samples were collected weekly, so they would better capture the variation of DMS oxidation at a finer time scale. If we average the  $SO_4^{2-}_{NSS}$  concentration in the ambient collector to the same sampling window as the marine collector, a similar seasonal trend was observed. Meantime, the anthropogenic flux ranged from near zero to ~2.5 nmol/m<sup>3</sup>, with little variation throughout the year and no clear seasonal trend. Overall, because of the low contribution of  $SO_4^{2-}_{anth}$ , the seasonal variation of sulfate concentration at Baring Head is mainly controlled by the variation of  $SO_4^{2-}_{DMS}$ .

However, uncertainties exist in our source appointment because (1) the sampling intervals were ~1 month for the marine collector and 1 week for the ambient collector, thus the marine collector could not record the accurate  $\delta^{34}S_{DMS}$  end-member; (2) the atmospheric chemistry of sulfate formation in the urban areas and the open ocean was different because of the differences in NO<sub>x</sub>, volatile organic carbon, and O<sub>3</sub> concentrations, hence the isotopic fractionation between SO<sub>2</sub> and sulfate may be different (Harris et al., 2012); and (3) the  $\delta^{34}S$  of anthropogenic sulfate could display a wider range (Calhoun et al., 1991; Norman et al., 1999; Wadleigh, 2004). Nevertheless, these uncertainties would not significantly impact our calculation.

### 4. Conclusion

We investigated the source distribution and seasonal variations of size-aggregated sulfate aerosols on a coastal site at Baring Head, New Zealand from marine and ambient collectors. We suggested that using total  $[Na^+]$  or  $[Mg^{2+}]$  instead of purely ionic  $[Mg^{2+}]$  to estimate the sea-salt sulfate concentration could exclude the error caused by high Na in filter blanks and Mg complexes in sea-salt aerosols. In our samples, coarse aerosol sulfate was dominated by  $SO_4^{2-}_{SS}$ , the variation of which was mainly determined by a combination of wind speed and wave height; fine aerosol sulfate was dominated by  $SO_4^{2-}_{NSS}$ . Therefore, the sulfate in the oceanic CCN was mainly controlled by  $SO_4^{2-}_{NSS}$ .

 $\delta^{34}S_{NSS}$  in both collectors provided a direct observation of the relative importance of  $SO_4^{2-}_{DMS}$  and  $SO_4^{2-}_{anth}$  in the midlatitude Southern Ocean. The  $SO_4^{2-}_{NSS}$  in marine collector was primarily (88–100%) of DMS origin. The concentrations of  $SO_4^{2-}_{DMS}$  showed significant seasonal variation, which allow us to attribute the observed high summer CCN level in the Southern Ocean troposphere to the elevated DMS emission. The ambient  $SO_4^{2-}_{NSS}$  displayed a lower and wider range of  $\delta^{34}S$  values, indicating a mixture between  $SO_4^{2-}_{DMS}$  and  $SO_4^{2-}_{anth}$  with the average anthropogenic contribution range between 21 and 27%. Our observations suggested a much lower  $SO_4^{2-}_{anth}$  background in the Southern Hemisphere than in the Northern Hemisphere.

#### Acknowledgments

We thank funding support from the National Science Foundation and Royal Society of New Zealand's EAPSI Program and NIWA SSIF Climate and Atmosphere programs. We thank Pam Rogers (GNS Science), Gordon Brailsford and Zoë Buxton, Mark Fisher, Meng Liu, Jiayu Ren, and members of NIWA for their diligent work in filter collection and sample preparation. Supporting data are included in the SI file; any additional data may be obtained from Jianghanyang Li (li2502@purdue.edu).

#### References

- Alexander, B., Park, R. J., Jacob, D. J., Li, Q. B., Yantosca, R. M., Savarino, J., et al. (2005). Sulfate formation in sea-salt aerosols: Constraints from oxygen isotopes. Journal of Geophysical Research, 110, D10307. https://doi.org/10.1029/2004JD005659
- Allen, A. G., Dick, A. L., & Davison, B. M. (1997). Sources of atmospheric methanesulphonate, non-sea-salt sulphate, nitrate and related species over the temperate South Pacific. *Atmospheric Environment*, *31*(2), 191–205. https://doi.org/10.1016/1352-2310(96)00194-X
- Amrani, A., Said-Ahmad, W., Shaked, Y., & Kiene, R. P. (2013). Sulfur isotope homogeneity of oceanic DMSP and DMS. Proceedings of the National Academy of Sciences of the United States of America, 110(46), 18,413–18,418. https://doi.org/10.1073/pnas.1312956110
- Andronache, C., Chameides, W. L., Davis, D. D., Anderson, B. E., Pueschel, R. F., Bandy, A. R., et al. (1997). Gas-to-particle conversion of tropospheric sulfur as estimated from observations in the western North Pacific during PEM-west B. Journal of Geophysical Research, 102(D23), 28,511–28,538. https://doi.org/10.1029/97JD01969
- Ayers, G. P., & Gras, J. L. (1991). Seasonal relationship between cloud condensation nuclei and aerosol methanesulphonate in marine air. Nature, 353(6347), 834–835. https://doi.org/10.1038/353834a0
- Barnes, I., Hjorth, J., Mihalapoulos, N., & Mihalopoulos, N. (2006). Dimethyl sulfide and dimethyl sulfoxide and their oxidation in the atmosphere.
- Bates, T. S., Lamb, B. K., Guenther, A., Dignon, J., & Stoiber, R. E. (1992). Sulfur emissions to the atmosphere from natural sources. Journal of Atmospheric Chemistry, 14(1–4), 315–337. https://doi.org/10.1007/BF00115242
- Boers, R., Ayers, G. P., & Gras, J. L. (1994). Coherence between seasonal variation in satellite-derived cloud optical depth and boundary layer CCN concentrations at a mid-latitude Southern Hemisphere station. *Tellus*, 46B, 123–131.
- Cainey, J. M., Ayers, G. P., Sievering, H., Gillett, R. W., & Hooper, M. A. (1999). The use of magnesium as a marker for sea salt at Cape Grim, Baseline Atmospheric Program—1996, 7–14.
- Calhoun, J. A., & Bates, T. S. (1989). Sulfur isotope ratios: Tracers of non-sea salt sulfate in the remote atmosphere, ACS Symposium Series. Washington, DC: American Chemical Society. https://doi.org/10.1021/bk-1989-0393.ch022

- Calhoun, J. A., Bates, T. S., & Charlson, R. J. (1991). Sulfur isotope measurements of submicrometer sulfate aerosol particles over the Pacific Ocean. *Geophysical Research Letters*, 18(10), 1877–1880. https://doi.org/10.1029/91GL02304
- Capaldo, K., Corbett, J. J., Kasibhatla, P., Fischbeck, P., & Pandis, S. N. (1999). Effects of ship emissions on sulphur cycling and radiative climate forcing over the ocean. *Nature*, 400(6746), 743–746. https://doi.org/10.1038/23438
- Clarke, A. D., Owens, S. R., & Zhou, J. (2006). An ultrafine sea-salt flux from breaking waves: Implications for cloud condensation nuclei in the remote marine atmosphere. *Journal of Geophysical Research*, 111, D06202. https://doi.org/10.1029/2005JD006565
- Dams, R., Rahn, K. A., & Winchester, J. W. (1972). Evaluation of filter materials and impaction surfaces for nondestructive neutron activation analysis of aerosols. *Environmental Science & Technology*, 6(5), 441–448. https://doi.org/10.1021/es60064a001
- Davy, P. K. (2007). Composition and sources of aerosol in the Wellington region of New Zealand (PhD thesis). School of Chemical and Physical Sciences. Wellington, Victoria University of Wellington.
- Davy, P. K., Ancelet, T., Trompetter, W. J., Markwitz, A., & Weatherburn, D. C. (2012). Composition and source contributions of air particulate matter pollution in a New Zealand suburban town. Atmospheric Pollution Research, 3(1), 143–147. https://doi.org/10.5094/APR.2012.014
- Davy, P. K., Markwitz, A., Trompetter, W. J., & Weatherburn, D. C. (2008). Source apportionment of airborne particles at Seaview, Lower Hutt, GNS Science Consultancy Report 2013/146 (p. 92). Lower Hutt, New Zealand.
- Dorling, S. R., Davies, T. D., & Pierce, C. E. (1992). Cluster analysis: A technique for estimating the synoptic meteorological controls on air and precipitation chemistry—Method and applications. Atmospheric Environment, Part A, General Topics, 26(14), 2575–2581. https://doi.org/ 10.1016/0960-1686(92)90110-7
- Faloona, I. (2009). Sulfur processing in the marine atmospheric boundary layer: A review and critical assessment of modeling uncertainties. Atmospheric Environment, 43(18), 2841–2854. https://doi.org/10.1016/j.atmosenv.2009.02.043
- Ghahremaninezhad, R., Norman, A.-L., Abbatt, J. P. D., Levasseur, M., & Thomas, J. L. (2016). Biogenic, anthropogenic and sea salt sulfate size-segregated aerosols in the Arctic summer. Atmospheric Chemistry and Physics, 16(8), 5191–5202. https://doi.org/10.5194/ acp-16-5191-2016
- Gong, S. L. (2003). A parameterization of sea-salt aerosol source function for sub- and super-micron particles. *Global Biogeochemical Cycles*, 17(4), 1097. https://doi.org/10.1029/2003GB002079
- Górka, M., Skrzypek, G., Hałas, S., Jędrysek, M. O., & Strąpoć, D. (2017). Multi-seasonal pattern in 5-year record of stable H, O and S isotope compositions of precipitation (Wrocław, SW Poland). Atmospheric Environment, 158, 197–210. https://doi.org/10.1016/j.atmosenv.2017.03.033
- Harris, E., Sinha, B., Hoppe, P., Foley, S., & Borrmann, S. (2012). Fractionation of sulfur isotopes during heterogeneous oxidation of SO<sub>2</sub> on sea salt aerosol: A new tool to investigate non-sea salt sulfate production in the marine boundary layer. *Atmospheric Chemistry and Physics*, 12(10), 4619–4631. https://doi.org/10.5194/acp-12-4619-2012
- Harris, E., Sinha, B., Hoppe, P., & Ono, S. (2013). High-precision measurements of <sup>33</sup>S and <sup>34</sup>S fractionation during SO<sub>2</sub> oxidation reveal causes of seasonality in SO<sub>2</sub> and sulfate isotopic composition. *Environmental Science & Technology*, 47(21), 12,174–12,183. https://doi.org/ 10.1021/es402824c
- Hill-Falkenthal, J., Priyadarshi, A., & Thiemens, M. (2012). Differentiating sulfate aerosol oxidation pathways for varying source altitudes using  $^{35}$ S and  $\Delta^{17}$ O tracers. *Journal of Geophysical Research*, *117*, D18302. https://doi.org/10.1029/2012JD018242
- Huang, S., Poulain, L., van Pinxteren, D., van Pinxteren, M., Wu, Z., Herrmann, H., & Wiedensohler, A. (2017). Latitudinal and seasonal distribution of particulate MSA over the Atlantic using a validated quantification method with HR-ToF-AMS. *Environmental Science & Technology*, *51*(1), 418–426. https://doi.org/10.1021/acs.est.6b03186
- Hudson, J. G., & Noble, S. (2006). Cloud condensation nuclei sizes, in AGU Fall Meeting Abstracts.
- I. E. A. (2014). CO<sub>2</sub> emissions from fuel combustion 2014.
- I. P. C. C. (2007). The Fourth Assessment Report of the Intergovernmental Panel on Climate Change. Geneva, Switzerland.
- Jaeglé, L., Quinn, P. K., Bates, T. S., Alexander, B., & Lin, J.-T. (2011). Global distribution of sea salt aerosols: New constraints from in situ and remote sensing observations. *Atmospheric Chemistry and Physics*, *11*(7), 3137–3157. https://doi.org/10.5194/acp-11-3137-2011
- Jayarathne, T., Sultana, C. M., Lee, C., Malfatti, F., Cox, J. L., Pendergraft, M. A., et al. (2016). Enrichment of saccharides and divalent cations in sea spray aerosol during two phytoplankton blooms. *Environmental Science & Technology*, 50(21), 11,511–11,520. https://doi.org/10.1021/acs.est.6b02988
- Jensen, D. R., Zeisse, C. R., Littfin, K. M., & Gathman, S. G. (1997). EOPACE (Electro-Optical Propagation Assessment in Coastal Environments): Overview and initial accomplishments, in propagation and imaging through the atmosphere. *International Society for Optics and Photonics*, 3125, 98–109.
- Keene, W. C., Maring, H., Maben, J. R., Kieber, D. J., Pszenny, A. A. P., Dahl, E. E., et al. (2007). Chemical and physical characteristics of nascent aerosols produced by bursting bubbles at a model air-sea interface. *Journal of Geophysical Research*, 112, D21202. https://doi.org/10.1029/ 2007JD008464
- Keene, W. C., Pszenny, A. A. P., Galloway, J. N., & Hawley, M. E. (1986). Sea-salt corrections and interpretation of constituent ratios in marine precipitation. Journal of Geophysical Research, 91(D6), 6647. https://doi.org/10.1029/JD091iD06p06647
- Korhonen, H., Carslaw, K. S., Spracklen, D. V., Mann, G. W., & Woodhouse, M. T. (2008). Influence of oceanic dimethyl sulfide emissions on cloud condensation nuclei concentrations and seasonality over the remote Southern Hemisphere oceans: A global model study. *Journal* of Geophysical Research, 113, D15204. https://doi.org/10.1029/2007JD009718
- Krouse, H. R., & Grinenko, V. A. (1991). Stable isotopes: natural and anthropogenic sulphur in the environment.
- Kulmala, M., Pirjola, L., & Makela, J. M. (2000). Stable sulphate clusters as a source of new atmospheric particles. *Nature*, 404(6773), 66–69. https://doi.org/10.1038/35003550
- Lana, A., Bell, T. G., Simó, R., Vallina, S. M., Ballabrera-Poy, J., Kettle, A. J., et al. (2011). An updated climatology of surface dimethlysulfide concentrations and emission fluxes in the global ocean. *Global Biogeochemical Cycles*, 25, GB1004. https://doi.org/10.1029/ 2010qb003850
- Law, C. S., Smith, M. J., Harvey, M. J., Bell, T. G., Cravigan, L. T., Elliott, F. C., et al. (2017). Overview and preliminary results of the Surface Ocean Aerosol Production (SOAP) campaign. Atmospheric Chemistry and Physics, 17(22), 13,645–13,667. https://doi.org/10.5194/ acp-17-13645-2017
- Lewis, E. R., & Schwartz, S. E. (2004). Sea salt aerosol production: Mechanisms, methods, measurements, and models—A critical review. In *Geophysical Monograph Series* (Vol. 152). Washington, DC: American Geophysical Union. https://doi.org/10.1029/GM152
- Lin, M., Biglari, S., Zhang, Z., Crocker, D., Tao, J., Su, B., et al. (2017). Vertically uniform formation pathways of tropospheric sulfate aerosols in East China detected from triple stable oxygen and radiogenic sulfur isotopes. *Geophysical Research Letters*, 44, 5187–5196. https://doi.org/ 10.1002/2017GL073637
- Mårtensson, M., Nilsson, E. D., De Leeuw, G., Cohen, L. H., & Hansson, H. C. (2003). Laboratory simulations of the primary marine aerosol generated by bubble bursting. *JGR*, *108*(D9), 4297. https://doi.org/10.1029/2002JD002263

- McArdle, N., Liss, P., & Dennis, P. (1998). An isotopic study of atmospheric sulphur at three sites in Wales and at Mace Head, Eire. Journal of Geophysical Research, 103(D23), 31,079–31,094. https://doi.org/10.1029/98JD01664
- McCoy, D. T., Burrows, S. M., Wood, R., Grosvenor, D. P., Elliott, S. M., Ma, P.-L., et al. (2015). Natural aerosols explain seasonal and spatial patterns of Southern Ocean cloud albedo. *Science Advances*, 1(6), e1500157. https://doi.org/10.1126/sciadv.1500157

McInnes, L. M., Covert, D. S., Quinn, P. K., & Germani, M. S. (1994). Measurements of chloride depletion and sulfur enrichment in individual sea-salt particles collected from the remote marine boundary layer. *Journal of Geophysical Research*, 99(D4), 8257–8268. https://doi.org/ 10.1029/93JD03453

Ministry of the Environment (2004). New Zealand, emission inventories for CO, NO<sub>x</sub>, SO<sub>2</sub>, ozone, benzene and benzo(a)pyrene in New Zealand. Air Quality Technical Report 44.

Mizutani, Y., & Rafter, T. A. (1969). Oxygen isotopic composition of sulfates. Part 5. Isotopic composition of sulfate in rain water. Gracefield, New Zealand., Lower Hutt, New Zealand: Inst. of Nuclear Sciences.

Monahan, E. C., Spiel, D. E., & Davidson, K. L. (1986). A model of marine aerosol generation via whitecaps and wave disruption. In D. Reidel (Ed.), Oceanic whitecaps: And their role in air-sea exchange processes (pp. 167–174). New York: Springer.

Murphy, D. M., Anderson, J. R., Quinn, P. K., McInnes, L. M., Brechtel, F. J., Kreidenweis, S. M., et al. (1998). Influence of sea-salt on aerosol radiative properties in the Southern Ocean marine boundary layer. *Nature*, 392(6671), 62–65. https://doi.org/10.1038/32138

Nielsen, H. (1974). Isotopic composition of the major contributors to atmospheric sulfur. *Tellus*, 26(1–2), 213–221.

Norman, A. L., Barrie, L. A., Toom-Sauntry, D., Sirois, A., Krouse, H. R., Li, S. M., & Sharma, S. (1999). Sources of aerosol sulphate at Alert: Apportionment using stable isotopes. *Journal of Geophysical Research*, 104(D9), 11,619–11,631. https://doi.org/10.1029/1999JD900078 Novák, M., Jacková, I., & Prechová, E. (2001). Temporal trends in the isotope signature of air-borne sulfur in Central Europe. *Environmental* 

Science & Technology, 35(2), 255–260. https://doi.org/10.1021/es0000753 O'Dowd, C. D., De Leeuw, G., O'Dowd, C. D., & De Leeuw, G. (2007). Marine aerosol production: A review of the current knowledge.

O'Dowd, C. D., De Leeuw, G., O'Dowd, C. D., & De Leeuw, G. (2007). Marine aerosol production: A review of the current knowledge. Philosophical Transactions of The Royal Society London A Mathematical Physical and Engineering Sciences, 365(1856), 1753–1774. https:// doi.org/10.1098/rsta.2007.2043

O'Dowd, C. D., & Smith, M. H. (1993). Physicochemical properties of aerosols over the northeast Atlantic: Evidence for windspeed-related submicron sea-salt aerosol production. Journal of Geophysical Research, 98(D1), 1137–1149. https://doi.org/10.1029/92JD02302

- O'Dowd, C. D., Smith, M. H., Consterdine, I. E., & Lowe, J. A. (1997). Marine aerosol, sea-salt, and the marine sulphur cycle: A short review. Atmospheric Environment, 31(1), 73–80.
- Oduro, H., Van Alstyne, K. L., & Farquhar, J. (2012). Sulfur isotope variability of oceanic DMSP generation and its contributions to marine biogenic sulfur emissions. *Proceedings of the National Academy of Sciences of the United States of America*, 109(23), 9012–9016. https://doi. org/10.1073/pnas.1117691109

Patris, N., Delmas, R., Legrand, M., De Angelis, M., Ferron, F. A., Stiévenard, M., & Jouzel, J. (2002). First sulfur isotope measurements in central Greenland ice cores along the preindustrial and industrial periods. *Journal of Geophysical Research*, 107(D11), 4115. https://doi.org/ 10.1029/2001JD000672

Patris, N., Mihalopoulos, N., Baboukas, E. D., & Jouzel, J. (2000). Isotopic composition of sulfur in size-resolved marine aerosols above the Atlantic Ocean. *Journal of Geophysical Research*, 105(D11), 14,449–14,457. https://doi.org/10.1029/1999JD901101

Paxian, A., Eyring, V., Beer, W., Sausen, R., & Wright, C. (2010). Present-day and future global bottom-up ship emission inventories including polar routes. *Environmental Science & Technology*, 44(4), 1333–1339. https://doi.org/10.1021/es9022859

Priyadarshi, A., Hill-Falkenthal, J., Coupal, E., Dominguez, G., & Thiemens, M. H. (2012). Measurements of <sup>35</sup>S in the marine boundary layer at La Jolla, California: A new technique for tracing air mass mixing during Santa Ana events. *Journal of Geophysical Research*, *117*, D08301. https://doi.org/10.1029/2011JD016878

Proemse, B. C., & Mayer, B. (2012). Tracing industrial nitrogen and sulfur emissions in the Athabasca oil sands region using stable isotopes. Developments in Environmental Science, 11, 243–266. https://doi.org/10.1016/B978-0-08-097760-7.00011-1

Quinn, P. K., Bates, T. S., Schulz, K., & Shaw, G. E. (2009). Decadal trends in aerosol chemical composition at Barrow, Alaska: 1976–2008. Atmospheric Chemistry and Physics, 9(22), 8883–8888. https://doi.org/10.5194/acp-9-8883-2009

Rees, C. E., Jenkins, W. J., & Monster, J. (1978). The sulphur isotopic composition of ocean water sulphate. *Geochimica et Cosmochimica Acta*, 42(4), 377–381. https://doi.org/10.1016/0016-7037(78)90268-5

Rempillo, O., Seguin, A. M., Norman, A.-L., Scarratt, M., Michaud, S., Chang, R., et al. (2011). Dimethyl sulfide air-sea fluxes and biogenic sulfur as a source of new aerosols in the Arctic fall. *Journal of Geophysical Research*, 116, D00S04. https://doi.org/10.1029/ 2011JD016336

Sakai, H., Casadevall, T. J., & Moore, J. G. (1982). Chemistry and isotope ratios of sulfur in basalts and volcanic gases at Kilauea volcano, Hawaii. Geochimica et Cosmochimica Acta, 46(5), 729–738. https://doi.org/10.1016/0016-7037(82)90024-2

Sanusi, A., Norman, A., Burridge, C., Wadleigh, M., & Tang, W. (2006). Determination of the S isotope composition of methanesulfonic acid. Analytical Chemistry, 78(14), 4964–4968. https://doi.org/10.1021/AC0600048

Savoie, D. L., & Prospero, J. M. (1989). Comparison of oceanic and continental sources of non-sea-salt sulphate over the Pacific Ocean. *Nature*, 339(6227), 685–687. https://doi.org/10.1038/339685a0

Seguin, A. M., Norman, A.-L., Eaton, S., & Wadleigh, M. (2011). Seasonality in size segregated biogenic, anthropogenic and sea salt sulfate aerosols over the North Atlantic. *Atmospheric Environment*, *45*(38), 6947–6954. https://doi.org/10.1016/j.atmosenv.2011.09.033

Sievering, H., Cainey, J., Harvey, M., McGregor, J., Nichol, S., & Quinn, P. (2004). Aerosol non-sea-salt sulfate in the remote marine boundary layer under clear-sky and normal cloudiness conditions: Ocean-derived biogenic alkalinity enhances sea-salt sulfate production by ozone oxidation. *Journal of Geophysical Research*, *109*, D19317. https://doi.org/10.1029/2003JD004315

Steinkamp, K., Mikaloff Fletcher, S. E., Brailsford, G., Smale, D., Moore, S., Keller, E. D., et al. (2017). Atmospheric CO<sub>2</sub> observations and models suggest strong carbon uptake by forests in New Zealand. *Atmospheric Chemistry and Physics*, 17(1), 47–76. https://doi.org/10.5194/ acp-17-47-2017

Udisti, R., Bazzano, A., Becagli, S., Bolzacchini, E., Caiazzo, L., Cappelletti, D., et al. (2016). Sulfate source apportionment in the Ny-Ålesund (Svalbard Islands) Arctic aerosol. *Rendiconti Lincei*, *27*(S1), 85–94. https://doi.org/10.1007/s12210-016-0517-7

Udisti, R., Dayan, U., Becagli, S., Busetto, M., Frosini, D., Legrand, M., et al. (2012). Sea spray aerosol in central Antarctica. Present atmospheric behaviour and implications for paleoclimatic reconstructions. *Atmospheric Environment*, 52, 109–120. https://doi.org/10.1016/ i.atmosenv.2011.10.018

Van Eijk, A. M. J., Kusmierczyk-Michulec, J. T., Francius, M. J., Tedeschi, G., Piazzola, J., Merritt, D. L., & Fontana, J. D. (2011). Sea-spray aerosol particles generated in the surf zone. Journal of Geophysical Research, 116, D19210. https://doi.org/10.1029/2011JD015602

Wadleigh, M. A. (2004). Sulphur isotopic composition of aerosols over the western North Atlantic Ocean. Canadian Journal of Fisheries and Aquatic Sciences, 61(5), 817–825. https://doi.org/10.1139/f04-073

Xu, G., & Gao, Y. (2015). Characterization of marine aerosols and precipitation through shipboard observations on the transect between 31°N-32°S in the West Pacific, Atmos. *Pollution Research*, 6(1), 154–161.

Yang, Y., Wang, H., Smith, S. J., Easter, R., Ma, P.-L., Qian, Y., et al. (2017). Global source attribution of sulfate concentration and direct and indirect radiative forcing. *Atmospheric Chemistry and Physics*, *17*(14), 8903–8922. https://doi.org/10.5194/acp-17-8903-2017

Zhu, G., Guo, Q., Chen, T., Lang, Y., Peters, M., Tian, L., et al. (2016). Chemical and sulfur isotopic composition of precipitation in Beijing, China. Environmental Science and Pollution Research, 23(6), 5507–5515. https://doi.org/10.1007/s11356-015-5746-2